%% if you are submitting an initial manuscript then you should have submission as an option here
%% if you are submitting a revised manuscript then you should have revision as an option here
%% otherwise options taken by the article class will be accepted
\documentclass[finalversion]{FPSAC2021}
\articlenumber{16}
%% but DO NOT pass any options (or change anything else anywhere) which alters page size / layout / font size etc

%% note that the class file already loads {amsmath, amsthm, amssymb}

\usepackage{tikz}

\newtheorem{thm}{Theorem}[section]
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{con}[thm]{Conjecture}

\def\beq{\begin{equation}}
\def\eeq{\end{equation}}

\def\ben{\begin{enumerate}}
\def\een{\end{enumerate}}

\def\ble{\begin{lem}}
\def\ele{\end{lem}}

\def\bco{\begin{cor}}
\def\eco{\end{cor}}

\def\bcon{\begin{con}}
\def\econ{\end{con}}

\def\bth{\begin{thm}}
\def\eth{\end{thm}}

\def\bprf{\begin{proof}}
\def\eprf{\end{proof}}

\newcommand{\dil}{\displaystyle}

\def\bfi{\begin{figure}\begin{center}}
\def\efi{\end{center}\end{figure}}

\newcommand{\eqqed}[1]{\vspace{5pt}$\rule{1ex}{0ex}\hfill{\dil#1}\hfill\qed$}
\newcommand{\qmq}[1]{\quad\mbox{#1}\quad}

\newcommand{\ree}[1]{(\ref{#1})}

\def\al{\alpha}
\def\be{\beta}
\def\ka{\kappa}
\renewcommand{\th}{\theta}

\def\capt{\caption}
\def\emp{\emptyset}
\def\gt{\rhd}
\def\gte{\unrhd}
\def\iso{\cong}
\def\lt{\lhd}
\def\lte{\unlhd}
\def\sbe{\subseteq}


\newcommand{\ce}[1]{\lceil{#1}\rceil}
\newcommand{\fl}[1]{\lfloor{#1}\rfloor}


\DeclareMathOperator{\cen}{cen}
\DeclareMathOperator{\core}{core}
\DeclareMathOperator{\GK}{GK}
\DeclareMathOperator{\mi}{mi}
\DeclareMathOperator{\rk}{rk}


\usepackage{lipsum}

%% define your title in the usual way
\title{On a rank-unimodality conjecture of Morier-Genoud and Ovsienko}

%% define your authors in the usual way
%% use \addressmark{1}, \addressmark{2} etc for the institutions, and use \thanks{} for contact details
\author[T. McConville, B. Sagan, C. Smyth]{Thomas McConville\addressmark{1}, Bruce Sagan\addressmark{2}, \and Clifford Smyth\addressmark{3}}

%% then use \addressmark to match authors to institutions here
\address{\addressmark{1}Department of Mathematics, Kennesaw State University \\ \addressmark{2}Department of Mathematics, Michigan State University \\ \addressmark{3}Department of Mathematics and Statistics, University of North Carolina at Greensboro}

%% put the date of submission here
\received{\today}

%% leave this blank until submitting a revised version
%\revised{}

%% put your English abstract here, or comment this out if you don't have one yet
%% please don't use custom commands in your abstract / resume, as these will be displayed online
%% likewise for citations -- please don't use \cite, and instead write out your citation as something like (author year)
\abstract{Let $\alpha=(a,b,\ldots)$ be a composition.  Consider the associated poset $F(\alpha)$, called a fence, whose covering relations are
\[
x_1\lhd x_2 \lhd \ldots\lhd x_{a+1}\rhd x_{a+2}\rhd \ldots\rhd x_{a+b+1}\lhd x_{a+b+2}\lhd \ldots\ .
\]
We study the associated distributive lattice $L(\alpha)$ consisting of all lower order ideals of $F(\alpha)$.  
These lattices are important in the theory of cluster algebras and their rank generating functions can be used to define $q$-analogues of rational numbers.
In particular, we make progress on a recent conjecture of Morier-Genoud and Ovsienko that $L(\alpha)$ is rank unimodal.  We show that if one of the parts of $\alpha$ is greater than the sum of the others, then the conjecture is true.  We conjecture that $L(\alpha)$ enjoys the stronger properties of having a nested chain decomposition and having a rank sequence which is either top or bottom interlacing, the latter being a recently defined property of sequences.  We verify that these properties hold for compositions with at most three parts and for what we call $d$-divided posets, generalizing work of Claussen and simplifying a construction of Gansner.}

%% put your French abstract here, or comment this out if you don't have one
\resume{Soit $\alpha=(a,b,\ldots)$ une composition.  Consid\'erons l'ensemble partiellement ordonn\'e associ\'e $F(\alpha)$, appelé une clôture, dont les r\'elations de couverture sont
\[
x_1\lhd x_2 \lhd \ldots\lhd x_{a+1}\rhd x_{a+2}\rhd \ldots\rhd x_{a+b+1}\lhd x_{a+b+2}\lhd \ldots\ .
\]
Nous \'etudions le treillis distributif associ\'e $L(\alpha)$ compos\'e de tous les id\'eaux inf\'erieurs de $F(\alpha)$.
Ces treillis sont importants en la th\'eorie d'alg\`ebres ammass\'ees et leurs fonctions g\'en\'eratrices de rang peuvent \^etre utilis\'ees
pour d\'efinir des $q$-analogues des nombres rationnels.  
En particulier, nous progressons sur une conjecture r\'ecente de Morier-Genoud et Ovsienko que $L(\alpha)$ est rang unimodal.
Nous v\'erifions la conjecture quand une des parties de la composition est plus grande que la somme des autres.
Nous conjecturons que $L(\alpha)$ a les propri\'et\'es plus puissantes d'avoir une d\'ecompoisition en cha\^{\i}nes imbriqu\'ees et d'avoir une s\'equence de rangs qui entrelace soit en haut soit en  bas, ce dernier \'etant une propri\'et\'e r\'ecemment d\'efinie.
Nous v\'erifions que ces propri\'et\'es tiennent pour les compositions avec au plus trois parties  en g\'en\'eralisant le travail de Claussen.}

%% put your keywords here, or comment this out if you don't have them yet
\keywords{heaviness, interlacing, distributive lattice, fence, nested chain decomposition, rank unimodal}

%% you can include your bibliography however you want, but using an external .bib file is STRONGLY RECOMMENDED and will make the editor's life much easier
%% regardless of how you do it, please use numerical citations, ie. [xx, yy] in the text

%% this sample uses biblatex, which (among other things) takes care of URLs in a more flexible way than bibtex
%% but you can use bibtex if you want
\usepackage[backend=bibtex]{biblatex}
\addbibresource{16McConville.bib}
%% note the \printbibliography command at the end of the file which goes with these biblatex commands

\begin{document}
\begin{thebibliography}{30}

%% \bibitem{ath:esl}
%% Christos~A. Athanasiadis.
%% \newblock Edgewise subdivisions, local {$h$}-polynomials, and excedances in the
%%   wreath product {$\mathbb{Z}_r \wr \mathfrak{S}_n$}.
%% \newblock {\em SIAM J. Discrete Math.}, 28(3):1479--1492, 2014.

%% \bibitem{ath:gcg}
%% Christos~A. Athanasiadis.
%% \newblock Gamma-positivity in combinatorics and geometry.
%% \newblock {\em S\'{e}m. Lothar. Combin.}, 77:Art. B77i, 64, [2016-2018].

%% \bibitem{BE:sbi}
%% Anders Bj\"{o}rner and Torsten Ekedahl.
%% \newblock On the shape of {B}ruhat intervals.
%% \newblock {\em Ann. of Math. (2)}, 170(2):799--817, 2009.

%% \bibitem{BJM:hpz}
%% Matthias Beck, Katharina Jochemko, and Emily McCullough.
%% \newblock {$h^\ast$}-polynomials of zonotopes.
%% \newblock {\em Trans. Amer. Math. Soc.}, 371(3):2021--2042, 2019.

%% \bibitem{bra:ulr}
%% Petter Br\"{a}nd\'{e}n.
%% \newblock Unimodality, log-concavity, real-rootedness and beyond.
%% \newblock In {\em Handbook of enumerative combinatorics}, Discrete Math. Appl.
%%   (Boca Raton), pages 437--483. CRC Press, Boca Raton, FL, 2015.

%% \bibitem{BS:sdr}
%% Petter Br\"{a}nd\'{e}n and Liam Solus.
%% \newblock Symmetric decompositions and real-rootedness.
%% \newblock Preprint {\texttt{arXiv:math.CO/1808.04141}}, 2018.

%% \bibitem{bre:lus}
%% Francesco Brenti.
%% \newblock Log-concave and unimodal sequences in algebra, combinatorics, and
%%   geometry: an update.
%% \newblock In {\em Jerusalem combinatorics '93}, volume 178 of {\em Contemp.
%%   Math.}, pages 71--89. Amer. Math. Soc., Providence, RI, 1994.

%% \bibitem{cla:epp}
%% Andrew Claussen.
%% \newblock {\em Expansion posets for polygon cluster algebras}.
%% \newblock PhD thesis, Michigan State University, 2020.

%% \bibitem{gan:loi}
%% Emden~R. Gansner.
%% \newblock On the lattice of order ideals of an up-down poset.
%% \newblock {\em Discrete Math.}, 39(2):113--122, 1982.

%% \bibitem{GK:ptt}
%% Curtis Greene and Daniel~J. Kleitman.
%% \newblock Proof techniques in the theory of finite sets.
%% \newblock In {\em Studies in combinatorics}, volume~17 of {\em MAA Stud.
%%   Math.}, pages 22--79. Math. Assoc. America, Washington, D.C., 1978.

%% \bibitem{GK:svs}
%% Curtis Greene and Daniel~J. Kleitman.
%% \newblock Strong versions of {S}perner's theorem.
%% \newblock {\em J. Combinatorial Theory Ser. A}, 20(1):80--88, 1976.

%% \bibitem{MGO:qdr}
%% Sophie Morier-Genoud and Valentin Ovsienko.
%% \newblock {$q$}-deformed rationals and {$q$}-continued fractions.
%% \newblock {\em Forum Math. Sigma}, 8:Paper No. e13, 55, 2020.

%% \bibitem{MSS:ruc}
%% Thomas McConville, Bruce Sagan, and Clifford Smyth.
%% \newblock On a rank-unimodality conjecture of Morier-Genoud and Ovsienko.
%% \newblock Preprint {\texttt{arXiv:math.CO/2008.13232}}, 2020.

%% \bibitem{MZ:rpl}
%% Emanuele Munarini and Norma Zagaglia~Salvi.
%% \newblock On the rank polynomial of the lattice of order ideals of fences and
%%   crowns.
%% \newblock {\em Discrete Math.}, 259(1-3):163--177, 2002.

%% \bibitem{MSW:pca}
%% Gregg Musiker, Ralf Schiffler, and Lauren Williams.
%% \newblock Positivity for cluster algebras from surfaces.
%% \newblock {\em Adv. Math.}, 227(6):2241--2308, 2011.

%% \bibitem{pro:cfp}
%% James Propp.
%% \newblock The combinatorics of frieze patterns and {M}arkoff numbers.
%% \newblock Preprint {\texttt{arXiv:math.CO/0511633}}, 2005.

%% \bibitem{sag:aoc}
%% Bruce~E. Sagan.
%% \newblock {\em Combinatorics: the {A}rt of {C}ounting}.
%% \newblock Graduate Studies in Mathematics. American Mathematical Society,
%%   Providence, RI, 2020, to appear.

%% \bibitem{sch:cef}
%% Ralf Schiffler.
%% \newblock A cluster expansion formula ({$A_n$} case).
%% \newblock {\em Electron. J. Combin.}, 15(1):Research paper 64, 9, 2008.

%% \bibitem{ST:caa}
%% Ralf Schiffler and Hugh Thomas.
%% \newblock On cluster algebras arising from unpunctured surfaces.
%% \newblock {\em Int. Math. Res. Not. IMRN}, (17):3160--3189, 2009.

%% \bibitem{sta:lus}
%% Richard~P. Stanley.
%% \newblock Log-concave and unimodal sequences in algebra, combinatorics, and
%%   geometry.
%% \newblock In {\em Graph theory and its applications: {E}ast and {W}est
%%   ({J}inan, 1986)}, volume 576 of {\em Ann. New York Acad. Sci.}, pages
%%   500--535. New York Acad. Sci., New York, 1989.

%% \bibitem{SV:uqi}
%% Jan Schepers and Leen Van~Langenhoven.
%% \newblock Unimodality questions for integrally closed lattice polytopes.
%% \newblock {\em Ann. Comb.}, 17(3):571--589, 2013.

%% \bibitem{sch:caa}
%% Ralf Schiffler.
%% \newblock On cluster algebras arising from unpunctured surfaces. {II}.
%% \newblock {\em Adv. Math.}, 223(6):1885--1923, 2010.

%% \bibitem{sol:sns}
%% Liam Solus.
%% \newblock Simplices for numeral systems.
%% \newblock {\em Trans. Amer. Math. Soc.}, 371(3):2089--2107, 2019.

%% \bibitem{yur:cef}
%% Toshiya Yurikusa.
%% \newblock Cluster expansion formulas in type {A}.
%% \newblock {\em Algebr. Represent. Theory}, 22(1):1--19, 2019.

%% \bibitem{yur:cce}
%% Toshiya Yurikusa.
%% \newblock Combinatorial cluster expansion formulas from triangulated surfaces.
%% \newblock {\em Electron. J. Combin.}, 26(2):Paper No. 2.33, 39, 2019.



%% \end{thebibliography}


\maketitle
%% note that you DO NOT have to put your abstract here -- it is generated by \maketitle and the \abstract and \resume commands above

\section{Basic definitions and background}

This extended abstract is a summary of the results in the paper of the same name and with the same authors~\cite{MSS:ruc}.

We will be studying the conjectured rank unimodality of certain distributive lattices.  We begin by defining the posets from which they arise.
Our terminology for partially ordered sets and other structures will follow Sagan's combinatorics text~\cite{sag:aoc}.
Let $\al=(\al_1,\al_2,\ldots,\al_s)$ be a {\em composition of $n-1$}, that is, a sequence of positive integers summing to $n-1$.  To simplify notation we will sometimes write $\al=(a,b,c,\ldots)$.
For each $\al$ we have a corresponding {\em fence poset}, $F=F(\al)$, with elements $x_1,x_2,\ldots,x_n$ and covering relations
\beq
\label{F(al)}
x_1\lt x_2 \lt \ldots\lt x_{a+1}\gt x_{a+2}\gt \ldots\gt x_{a+b+1}\lt x_{a+b+2}\lt\ldots\lt x_{a+b+c+1}\gt x_{a+b+c+2} \gt\ldots
\eeq
where $\lte$ is the order relation in $F$.
The Hasse diagram of the fence $F(2,3,1)$ is shown in Figure~\ref{F(2,3,1)}.
We will call the maximal chains of $F$ {\em segments} so that the $i$th part of $\al$ is equal to the length of the $i$th segment of $F$.  Because of this convention, the sum of the parts of $\al$ is one less than $\#F$, the cardinality of $F$.  %We will also denote cardinality by using the absolute value symbol.  

\bfi
\begin{tikzpicture}
\fill(0,1) circle(.1);
\fill(1,2) circle(.1);
\fill(2,3) circle(.1);
\fill(3,2) circle(.1);
\fill(4,1) circle(.1);
\fill(5,0) circle(.1);
\fill(6,1) circle(.1);
\draw (0,1)--(2,3)--(5,0)--(6,1);
\draw (0,.5) node{$x_1$};
\draw (1,1.5) node{$x_2$};
\draw (2,2.5) node{$x_3$};
\draw (3,1.5) node{$x_4$};
\draw (4,.5) node{$x_5$};
\draw (5,-.5) node{$x_6$};
\draw (6,.5) node{$x_7$};
\end{tikzpicture}
\capt{The fence $F(2,3,1)$ \label{F(2,3,1)}}
\efi

Given any poset $P$, its set of (lower) order ideals forms a distributive lattice $L(P)$.  We will shorten $L(F(\al))$ to $L(\al)$ and use similar abbreviations with other notation.  The complement of an order ideal of $P$ is an order ideal of $P^*$, the poset dual of $P$.  And 
if $\al=(\al_1,\ldots,\al_s)$ has an odd number of segments then $F(\al^r)\iso F(\al)^*$ where $\al^r=(\al_s,\ldots,\al_2,\al_1)$ is the reversal of $\al$ and  $\iso$ is a poset isomorphism.  Combining these two observations and translating to the corresponding lattices we have the following result which we record for future use.
\ble
\label{al^r}
For any $\al=(\al_1,\ldots,\al_s)$ with $s$ odd we have

\eqqed{
L(\al) \iso L(\al^r)^*.
}
\ele


The lattices $L(\al)$ will be our principal objects of study.  They are important objects in the theory of cluster algebras.  In particular, one can view $F(\al)$ as a quiver formed from the Dynkin diagram of type $A$ by replacing each cover $x\lt y$ with an arrow from $x$ to $y$.  
Then $L(\al)$ can be used to compute a mutation in a corresponding cluster algebra on a surface.  In fact, there are (at least) six different descriptions of $L(\al)$ or its dual which are useful for this computation.  These are in terms of perfect matchings on
snake graphs~\cite{pro:cfp}, 
perfect matchings of angles~\cite{yur:cef,yur:cce},
$T$-paths~\cite{sch:caa,sch:cef,ST:caa},
lattice paths on snake graphs~\cite{pro:cfp},
lattice paths of angles~\cite{cla:epp},
or $S$-paths~\cite{cla:epp}.

In order to introduce the conjecture on which we will focus, we need some definitions related to sequences and their generating functions.  We say that a sequence of nonnegative real numbers $a_0,a_1,\ldots,a_n$ is {\em unimodal} if there is some index $m$ such that
$$
a_0\le a_1\le \ldots \le a_m\ge a_{m+1}\ge \ldots \ge a_n.
$$
Unimodal sequences arise frequently in combinatorics, algebra, and geometry; see the 
survey articles of Stanley~\cite{sta:lus}, Brenti~\cite{bre:lus} or Br\"and\'en~\cite{bra:ulr}.  We will say that the generating function $f(q)=\sum_k a_k q^k$ has a property such as unimodality if its coefficient sequence does.

Now suppose that $P$ is a finite  poset.  We call $P$ {\em ranked} if, for all $x\in P$, the length of any saturated chain from a minimal element to $x$ is invariant.  This length is called the {\em rank of $x$} and denoted
$\rk x$.  We also define the {\em rank of $P$}, $\rk P$, to be the maximum of $\rk x$ over all $x\in P$. The {\em $k$th rank} of $P$ is the set
$$
R_j(P) = \{x\in P \mid \rk x=j\}
$$
and we let $r_j(P)=\#R_j(P)$.  Any finite distributive lattice is ranked by the cardinality of each element viewed as an order ideal of the corresponding poset of join irreducibles.  We say that $P$ is {\em rank unimodal} if the sequence $r_0(P), r_1(P),\ldots,r_n(P)$ is unimodal where $n=\rk P$.  We will similarly prepend ``rank" to other properties of sequences when applied to the rank sequence of a poset.  Our main object of study is the following conjecture of Morier-Genoud and Ovsienko.
\bcon[\cite{MGO:qdr}]
\label{MGO}
For any $\al$, the lattice $L(\al)$ is rank unimodal.
\econ

We note that Morier-Genoud and Ovsienko used the rank generating functions for the $L(\al)$ to define $q$-analogues for rational numbers.  Interestingly, special cases of this conjecture had already been proven even before it was stated because the problem is so natural in its own right.  Gansner~\cite{gan:loi} proved Conjecture~\ref{MGO} for certain dual fences which we call $d$-divided.  Munarini and Zagaglia~\cite{MZ:rpl} gave a different proof of the conjecture for $2$-divided fences which are those with 
$\al=(1,1,\ldots,1)$.  Since the conjecture was posed, Claussen~\cite{cla:epp} has shown that it is true for all fences with at most four segments.  One of our main results is that Conjecture~\ref{MGO} holds if one of the segments is sufficiently long.
\bth
\label{long}
Suppose $\al=(\al_1,\ldots,\al_s)$ and there is an index $t$ such that 
$$
\al_t > \sum_{i\neq t} \al_i.
$$
Then $L(\al)$ is rank unimodal.
\eth

We will also be interested in various strengthenings of Conjecture~\ref{MGO}.  To state them, we will need to define other properties of sequences and posets.  Say that the sequence $a_0,a_1,\ldots,a_n$ is {\em symmetric} if $a_k = a_{n-k}$ for $k<n/2$.  Symmetric unimodal sequences are common, for example a row of Pascal's triangle or the coefficients of a $q$-binomial coefficient.  Even when one does not have symmetry, there may be some relation between $a_k$ and $a_{n-k}$.  Call the sequence {\em top heavy} (respectively, {\em bottom heavy}) if $a_k\le a_{n-k}$ (respectively, $a_k\ge a_{n-k}$) for $k<n/2$.   As an illustration, a special case of a result of Bj\"orner and Ekedahl~\cite{BE:sbi} states that the rank sequence for Bruhat order on a crystallographic Coxeter group is top heavy.
More recently, a new property of sequences has been identified which implies both unimodality and heaviness.  Call the sequence {\em top interlacing} if
\beq
\label{ti}
a_0\le a_n \le a_1\le a_{n-1} \le \ldots\le a_{\ce{n/2}}
\eeq
where $\ce{\cdot}$ is the ceiling function.  Similarly, the sequence is {\em bottom interlacing} if
$$
a_n\le a_0 \le a_{n-1} \le a_1 \le \ldots \le a_{\fl{n/2}}
$$
with $\fl{\cdot}$ being the floor function.  See~\cite{ath:esl,ath:gcg,BJM:hpz,BS:sdr,SV:uqi,sol:sns} 
for research related to this concept.  We note that in the literature~\ree{ti} has been called ``alternately increasing''.  However, we prefer our terminology both because ``alternating'' usually refers to a sequence satisfying $a_0<a_1>a_2<a_3>\ldots$, and since~\ree{ti} implies that the first half of the sequence and the reverse of the second half interlace in the usual sense of the term.  We propose the following strengthening of Conjecture~\ref{MGO}.  In it, we refer to the rank sequence 
\beq
\label{r(al)}
r(\al) = (r_0(\al),r_1(\al),\ldots,r_n(\al))
\eeq
where $r_j(\al)=r_j(L(\al))$ and $n=\# F(\al)$.  
\bcon
\label{heavy}
Suppose $\al=(\al_1,\ldots,\al_s)$.
\ben
\item[(a)]  If $s=1$ then $r(\al) = (1,1,\ldots,1)$ is symmetric.
\item[(b)]  If $s$ is even, then $r(\al)$ is bottom interlacing.
\item[(c)]  Suppose $s\ge3$ is odd and let $\al'=(\al_2,\ldots,\al_{s-1})$.
	\ben
	\item[(i)] If $\al_1>\al_s$ then $r(\al)$ is bottom interlacing.
	\item[(ii)] If $\al_1<\al_s$ then $r(\al)$ is top interlacing.
	\item[(iii)] If $\al_1=\al_s$ then $r(\al)$ is symmetric, bottom interlacing, or top interlacing depending on whether 
	$r(\al')$ is symmetric, top interlacing, or bottom interlacing, respectively.
	\een
\een
\econ

Statement (a) in this conjecture is trivial, but is needed as a base case.  
We have verified this conjecture by computer for up to $5$ segments of lengths at most $10$, and for $6$ segments having lengths at most $5$. 
We have been able to prove the conjecture for various fences, including those with at most three segments and the $d$-divided posets, by showing that the corresponding lattices satisfy an even stronger condition which we now describe.

Let $P$ be a ranked poset with $\rk P = n$.  Also, let $C$ be a saturated $x$--$y$ chain in $P$.  The {\em center} and {\em interval} of $C$ are
the rational number
$$
\cen C =\frac{\rk x + \rk y}{2}
$$
and interval of integers
$$
[C]=[\rk x,\rk y],
$$
respectively.
A {\em chain decomposition} or {\em CD} of $P$ is a set partition of $P$ into saturated chains.  In a {\em symmetric chain decomposition} or {\em SCD}, every chain $C$ in the partition must satisfy $\cen C = n/2$.  Equivalently, if $C$ is an $x$--$y$ chain of the partition then  $\rk y = n-\rk x$.  If $P$ admits an SCD then its rank sequence is symmetric and unimodal.  In fact, $P$ even enjoys the strong Sperner property which says that, for all $k\ge1$,  the maximum cardinality of a union of $k$ antichains is just the sum of the $k$ largest ranks.  See the survey article of Greene and Kleitman~\cite{GK:ptt} for more information about chain decompositions and the Sperner property.

To deal with the case when the rank sequence is not symmetric,  consider a 
{\em nested chain composition}, NCD, which is a CD where any two of its chains $C,D$ satisfy either $[C]\sbe[D]$ or $[D]\sbe[C]$.  If $P$ admits an NCD then it is rank unimodal and still has the strong Sperner property.  We will be particularly concerned with a special type of NCD.  Call a CD {\em top centered} if every chain $C$ in the partition satisfies $\cen C =n/2$ or $\cen C = (n+1)/2$.  It follows easily  that this is an NCD and the rank sequence of $P$ is top interlacing.  
Similarly, a CD is {\em bottom centered} if its chains satisfy 
$\cen C = n/2$ or $(n-1)/2$.  Again, this is an NCD and the rank sequence is now bottom interlacing.  
Note also that if a poset has an NCD and its rank sequence is top or bottom interlacing then the NCD must be top or bottom centered, respectively.
This can be proven inductively using the observation that an NCD must contain a chain from a minimum rank element to a maximum rank element.
This leads to the strongest of our conjectures so far.
\bcon
\label{centered}
For any $\al$, the lattice $L(\al)$ admits a CD which is either symmetric, top centered, or bottom centered consistent with Conjecture~\ref{heavy}.
\econ

 We sketch proofs of a number of special cases of this conjecture in the sequel.  In particular, when the fence has at most three segments we have the following refinement of Claussen's result on the rank unimodality of the $L(\al)$.
\bth
\label{<=3}
If $\al$ has at most three parts then Conjecture~\ref{centered} is true.
\eth

The rest of this abstract is structured as follows.  In the next section we will sketch a proof of Theorem~\ref{long} in the case that the long segment is the first or the last.  We will do this using a recursion which will also permit us to replace the strict inequality with a weak one for these particular segments.  In Section~\ref{lms} we indicate how to complete the proof of Theorem~\ref{long}.  We will also describe an inductive procedure for proving Conjecture~\ref{centered} when a long segment exists, given that it holds for an appropriate base case.  The following section will be devoted to giving a construction to prove Theorem~\ref{<=3}.     %Using similar ideas we can construct a proof for $d$-divided posets which is modeled on, but simpler than, Gansner's demonstration.   Due to space limitations, we will not present that here.   %We end with a section outlining two more possible approaches to these conjectures.

\section{Long initial or final segments}
\label{lifs}

In this section we outline a proof of a stronger version of Theorem~\ref{long} where the long segment is either the first or the last.  This will be based on a recursion for the rank generating function
$$
r(q;\al) = \sum_{j\ge0} r_j(\al) q^j
$$
where the $r_j(\al)$ are given by~\ree{r(al)}.  The method of proof involves considering the ideals of $F(\al)$ which do or do not contain an element $x$, which we will call {\em toggling on $x$}.  Also, for the recursions to make sense, we must permit compositions
$(\al_1,\ldots,\al_s)$ where $\al_s=0$.  But in this case we just define
$$
F(\al_1,\ldots,\al_{s-1},0)=F(\al_1,\ldots,\al_{s-1}).
$$
\ble
\label{al_sRec}
Let $\al=(\al_1,\al_2,\ldots,\al_s)$.  Then for $s$ odd
$$
r(q;\al)=r(q;\al_1,\ldots,\al_{s-1},\al_s-1)  + q^{\al_s+1}\cdot r(q;\al_1,\ldots,\al_{s-2},\al_{s-1}-1)
$$
and for $s$ even
$$
r(q;\al) = r(q;\al_1,\ldots,\al_{s-2},\al_{s-1}-1) +  q\cdot r(q;\al_1,\ldots,\al_{s-1},\al_s-1).
$$
\ele


In order to make use of this lemma, we will have to consider the indices where the coefficients of a polynomial achieve their maximum.
Given $f=\sum_k a_k q^k$ we define the {\em set of maxima indices} as
$$
\mi(f) = \{k \mid a_k = m\}
$$
where $m$ is the maximum value of a coefficient of $f$.  Note that if $f$ is unimodal then $\mi(f)$ will be an interval of integers.  The next result is easy to prove.
\ble
\label{mi}
Let $f,g$ be unimodal polynomials and suppose that $\mi(f)\cap\mi(g) \neq\emp$.  Then $f+g$ is unimodal and 
$\mi(f+g)=\mi(f)\cap \mi(g)$.
\ele

The two preceding results can be used to prove a stronger version of Theorem~\ref{long} for the first segment.
\bth
\label{al_1}
If $\al=(\al_1,\al_2,\ldots,\al_s)$ satisfies
$$
\al_1\ge \al_2+\al_3+\cdots+\al_s
$$
then $r(q;\al)$  is unimodal with
$$
\mi(r(q;\al))=[\al_2+\al_3+\cdots+\al_s,\ \al_1].
$$
\eth


We have the same result for the last segment.
\bth
\label{al_s}
If $\al=(\al_1,\al_2,\ldots,\al_s)$ satisfies
$$
\al_s\ge \al_1+\al_2+\cdots+\al_{s-1}
$$
then $r(q;\al)$  is unimodal.
\eth


\section{Arbitrary long segments}
\label{lms}

Theorem 1 can be proved by using induction on the length of the long segment and the following lemma which locates the ranks of maximum size in the lattice coming from a poset with a long segment.
\ble
\label{MaxRk}
Let $\al=(\al_1,\ldots,\al_s)$ and $n=\#F(\al) $.  Suppose that for some $t$ we have
\beq
\label{al_t}
\al_t> \sum_{i\neq t} \al_i.
\eeq
Then the maximum size of a rank of $L=L(\al)$ is $\ell=\#L(F')$ where $F'$ is the poset obtained by removing the elements of segment $t$ from $F=F(\al)$. And this maximum occurs at ranks $m+1$ through $n-m-1$ where $m=\#F'$.
\ele



We now give an inductive method for proving that Conjecture~\ref{centered} holds.  To do so, we must first investigate the finer structure of $L(\al)$ where $\al$ has a long segment.  For any ranked poset $P$, let 
$$
P_k =\{x\in P \mid \rk x \le k\}
$$
and
$$
P^k =\{x\in P \mid \rk x \ge k\}.
$$
\ble
\label{iso}
Let $\al=(\al_1,\ldots,\al_s)$, $F=F(\al)$, and $L=L(\al)$.  Also let $n=\#F$ and $m=\#F'$ where $F'$ is as in Lemma~\ref{MaxRk}.  Suppose that for some $t$ we have
$$
\al_t> \sum_{i\neq t} \al_i.
$$
Let $G=F(\be)$ and $M=L(\be)$ where
$$
\be=(\al_1,\ldots,\al_{t-1},\al_t+1,\al_{t+1},\ldots,\al_s).
$$
Then we have isomorphisms
$$
L_{n-m-1}\iso M_{n-m-1}\qmq{and} L^{n-m}\iso M^{n-m+1}.
$$
\ele


We now have everything in place to state our inductive criterion for checking whether a poset with a long segment satisfies Conjecture~\ref{centered}.
\bth
Assume the hypotheses and notation of Lemma~\ref{iso}.  If $L$ has an NCD  then so does $M$.
Furthermore, if the NCD of $L$ is symmetric, top centered, or bottom centered then the NCD of $M$ has the same property.
\eth
%% \bprf
%% Suppose we have such an NCD of $L$.  We will keep the notation of the proof of Lemma~\ref{iso}.  Then, by this lemma, this NCD lifts to NCDs of $M_{n-m-1}$ and $M^{n-m+1}$.  What remains is to show how to connect these two NCDs through rank $n-m$ of $M$ to form an NCD of the whole poset.  
%% Let $R$, $R'$, and $R''$ denote ranks $n-m-1$ , $n-m$, and $n-m+1$ of $M$, respectively.  Note that by Lemma~\ref{MaxRk} we have $\#R=\#R'=\ell$.  As noted in the previous proof, no element of $R$ contains the coatom of the long
%% sement, $S$.  So there is a perfect matching from $J\in R$ to $J'\in R'$ by letting $J'=J\cup\{a\}$ where
%% where $a=\min(S-J)$.  To find the chains which continue to $R''$, consider $I=f(J)$ to
%%  see if it is connected in the NCD of $L$ to some $I''$ in the next rank up.  If so, then $J'$ will be covered by $J''=g^{-1}(I'')$  and so the sequence $J,J',J''$ will serve to connect the corresponding chains in $M$.  
%% It is easy to see that this construction will preserve whether the NCD is symmetric, top centered, or bottom centered which
%% completes the proof.
%% \eprf

Using the previous lemma and induction on the length of the long segment, we immediately get the following result.  Note that showing one lattice has an NCD of a certain form immediately gives an infinite family of lattices with NCDs of the same form.
\bco
Let $\al=(\al_1,\ldots,\al_s)$ where
\beq
\label{plus1}
\al_t= 1 + \sum_{i\neq t} \al_i
\eeq
for some $t$.  If $L=L(\al)$ has an NCD  then so does $M=L(\be)$ where
$$
\be=(\al_1,\ldots,\al_{t-1},\al_t+a,\al_{t+1},\ldots,\al_s)
$$
for any $a\ge 0$.  Furthermore, if the NCD of $L$ is symmetric, top centered, or bottom centered then the NCD of $M$ has the same property. 
\eco

Now one can use a computer to verify the following.
\bco
Let $F=F(\al_1,\ldots,\al_s)$ where
$$
\al_t> \sum_{i\neq t} \al_i
$$
where
$$
\sum_{i\neq t} \al_i \le 5.
$$
Then $L=L(\al)$ satisfies Conjecture~\ref{centered}.
\eco



\section{At most three segments}
\label{amts}

This section is devoted to sketching a proof of Theorem~\ref{<=3}.  The result is trivial if the fence $F(\al)$ has one segment  and fairly easy if it has two.  But for three, we will have to use a modified version of the famous Greene--Kleitman symmetric chain decomposition of the Boolean algebra of all subsets of a finite set~\cite{GK:svs}.

In particular, we will use the idea of a Greene--Kleitman core.  To define this object, let 
$w=w_1 w_2\ldots w_n$ be a sequence (or word) of zeros and ones.  The {\em Greene--Kleitman (GK) core of $w$}, $\GK(w)$, 
is a set of pairs of indices formed as follows.  If $w_i=0$ and $w_{i+1}=1$ then $(i,i+1)\in\GK(w)$.  We continue to add pairs $(i,j)$ to 
$\GK(w)$ as long as $w_i=0$ and $w_j=1$, where $i<j$ and all the elements between $w_i$ and $w_j$ are already in pairs of the core.  For example, if
$$
w= 110001011000111
$$
then
$$
\GK(w) = * * *\ \widehat{0\widehat{01}\widehat{01}1}* \widehat{0\widehat{01}1}\ *
$$
where elements not in the GK core have been replaced by stars, and pairs in the core are indicated by the hats.  Writing out the pairs themselves gives
$$
\GK(w) = \{(4,9),\ (5,6),\ (7,8),\ (11,14),\ (12,13)\}. 
$$
We will refer to the elements in the pairs of $\GK(w)$ as {\em matched}.  

To apply this idea to fences, we will have to modify the GK core.  And to do that it will be convenient to think of a fence $F$ as a partial order on $[n]=\{1,2,\ldots,n\}$ where, as usual, $n=\#F$.  When doing this, it will be important to distinguish $i\le j$ which is the usual  total order on the integers and $i\lte j$ which will be an order relation in $F$.  So consider $F=F(a,b,c)$ as the fence with covering relations
$$
b+c+1 \lt b+c+2 \lt \ldots \lt a+b+c+1 \gt b \gt b-1 \gt \ldots \gt 1 \lt b+1 \lt b+2 \lt \ldots \lt b+c.
$$
In other words label the second segment except for its maximum element with the elements of the interval $[1,b]$ from bottom to top.  Then label the elements of the third segment (except its minimum which has already been labeled) bottom to top with $[b+1,b+c]$.  Finally label the complete first segment with $[b+c+1,a+b+c+1]$ again from bottom to top.  Note that this labeling is a linear extension of $F$.  This labeling for the fence $F(2,3,1)$ is showing in Figure~\ref{labeling}.  

\bfi
\begin{tikzpicture}
\fill(0,1) circle(.1);
\fill(1,2) circle(.1);
\fill(2,3) circle(.1);
\fill(3,2) circle(.1);
\fill(4,1) circle(.1);
\fill(5,0) circle(.1);
\fill(6,1) circle(.1);
\draw (0,1)--(2,3)--(5,0)--(6,1);
\draw (0,.5) node{$5$};
\draw (1,1.5) node{$6$};
\draw (2,2.5) node{$7$};
\draw (3,1.5) node{$3$};
\draw (4,.5) node{$2$};
\draw (5,-.5) node{$1$};
\draw (6,.5) node{$4$};
\end{tikzpicture}
\capt{The labeling of $F(2,3,1)$ \label{labeling}}
\efi



Now associate with any subset $S\sbe F$ a word $w=w_S=w_1\ldots w_n$ where $w_i$ is one or zero depending on whether $i\in S$ or $i\not\in S$, respectively, and $n=\#F$.  Suppose $I\sbe F$ is an ideal and $w=w_I$.  
Since specifying $I$ and specifying $w$ are equivalent, we will often go back and forth without mention.

Given an ideal, $I$, call an element $f\in F$ {\em frozen} if it is unmatched in $\GK(w)$ and 
there exists $(i,j)\in GK(w)$ with $f \gt i$ or $f\lt j$.  Note that since $(i,j)$ is in the GK core we must have $i\not\in I$ and $j\in I$.  So 
$f\gt i$ implies $f\not\in I$, since $I$ is an ideal, and $w_f=0$.  Similarly, if $f\lt j$ then $f\in I$ and $w_f=1$.  
Whether an element is frozen or not depends upon the ideal under consideration, but we will make sure $I$ is clear from context.
Finally, define the 
{\em core} of $w=w_I$ by
$$
\core w = \GK(w) \cup \{f\in F \mid \text{$f$ is frozen}\}.
$$
We say that elements of $F$ not in $\core w$ are {\em free}.
For example, suppose $F=F(2,3,1)$ and $I=\{1,4,5\}$.  Then we first compute the GK core as indicated by the hats in
$$
w= 1 \widehat{0 \widehat{0 1} 1} 0 0.
$$ 
Since $(3,4)\in \GK(w)$ and $1\lt 4$ we have that $1$ is frozen in $w$.
Similarly, $(3,4)\in\GK(w)$ and $7\gt 3$ implies that $7$ is frozen.  One can also check that $6$ is free so that
$$
\core w = 1 \widehat{0 \widehat{0 1} 1} * 0
$$
or
$$
\core w = \{1,\ (2,5),\ (3,4),\ 7\}.
$$

We now complete the proof of Theorem~\ref{<=3} by forming, for each possible $\ka=\core w_I$ for some ideal $I$, a saturated chain containing $I$ as follows.  We start with $\overline{w}$ which is the word with every element not in $\core w$ equal to zero.  We then turn the zeros in $\overline{w}$ to $1$s one at a time from left to right until all free positions equal $1$ to form a saturated chain $C_\ka$.  In our running example, the chain would be
$$
C_\ka = \{ 1001100,\ 1001110\}.
$$
One can prove that in a fence $F(a,b,c)$ with $a> c$ that these form a bottom centered CD.  The case $a=c$ is dealt with similarly, and the case $a<c$ by Lemma~\ref{al^r}.




%% \section{Two more approaches}
%% \label{tma}

%% We end with two other possible methods for proving the conjectures in this abstract.  While we have not been able to make them bear fruit, we hope that someone else will be more successful.

%% \medskip

%% {\bf 1.  An explicit formula.}  There is an explicit formula for $r(q;\al)$ in terms of powers of $q$ and $q$-integers 
%% $[n]_q=1+q+\cdots + q^{n-1}$ where $n\ge0$.  This expression appeared in a more complicated form in Claussen's thesis~\cite{cla:epp} and formed part of the basis for his proof that Conjecture~\ref{MGO} is true for any fence with at most four segments.  It is easy to prove by toggling in a manner similar to that of the demonstration of Lemma~\ref{al_sRec}.  But now one toggles on subsets of 
%% \beq
%% \label{Z(al)}
%% Z(\al)=\{z_1,z_2,\ldots,z_u\}
%% \eeq
%% which is the set  of  maxima of $F(\al)$ written in order from left to right.  We will state the theorem without proof since the demonstration does not contain any new ideas.  A similar result holds for an even number of segments.
%% \bth[\cite{cla:epp}]
%% Let $\al=(\al_1,\ldots,\al_{2u-1})$ with $Z(\al)$ given by~\ree{Z(al)}.  Then
%% $$
%% r(q;\al) = \sum_{Z\sbe Z(\al)}  q^{\#Z} r(Z,1) r(Z,2) \cdots r(Z,u)
%% $$
%% where 
%% $$
%% r(Z,1) =
%% \begin{cases}
%% q^{\al_1} & \mbox{if $z_1\in Z$,}\\
%% [\al_1+1]_q & \mbox{if $z_1\not\in Z$,}
%% \end{cases}
%% $$
%% and
%% $$
%% r(Z,i)=
%% \begin{cases}
%% q^{\al_{2i-2}+\al_{2i-1}-1} & \mbox{if $z_{i-1},z_i\in Z$,}\\
%% q^{\al_{2i-2}}[\al_{2i-1}]_q & \mbox{if $z_{i-1}\in Z$ and $z_i\not\in Z$,}\\
%% q^{\al_{2i-1}}[\al_{2i-2}]_q & \mbox{if $z_{i-1}\not\in Z$ and $z_i\in Z$,}\\
%% 1+ q[\al_{2i-2}]_q[\al_{2i-1}]_q &\mbox{if $z_{i-1},z_i\not\in Z$.}
%% \end{cases}
%% $$
%% for $i>1$.
%% \eth

%% One can now write out $r(q;\al)$ for any desired $\al$ and try to use inductive arguments on these expressions.  It is easy to prove that
%% multiplying by $[n]_q$ preserves the interlacing conditions.  Unfortunately, addition is another matter and this is where the difficulty lies.

%% \medskip

%% {\bf 2.  Lexicographic CDs.}  There are two hurdles to using the Greene--Kleitman approach to constructing NCDs.  The first is finding a suitable labeling of the poset.  The other is coming up with an appropriate modification of $\GK(w)$ when $w$ is taken with respect to the labeling.  Here we propose another method which, given any labeling of any finite labeled poset $P$, produces a chain decomposition of the lattice of ideals $L=L(P)$ which may or may not be nested.

%% We construct the chains $C_1, C_2, C_3,\ldots$ of the CD  as follows.  Suppose $C_1,\ldots,C_{i-1}$ have been constructed. 
%% Since $P=[n]$ as sets, we can consider any ideal $I$ of $P$ as a subset of $\{1,\ldots,n\}$ and we will not make any distinction between an ideal and its subset.  So given two ideals, we can compare them in the lexicographic order on subsets.  Now we form $C_i$ by starting with the unique ideal $I_0$ which has minimum rank and is also lexicographically least among all elements of $L'=L-(C_1\cup\cdots\cup C_{i-1})$.
%% We now consider all ideals of $L'$ which cover $I_0$ and take the lexicographically least of them to be the next element $I_1$ on $C_i$.  We continue in this manner until we come to an ideal which has no cover in $L'$ at which point $C_i$ terminates.  We have the following conjecture which we have verified  for all fences $F$ with $\#F\le 7$.
%% \bcon
%% For every $\al$ there is a labeling of $F(\al)$ such that the correponding lexicographic CD is an NCD.
%% \econ

%% There are a couple of stumbling blocks we have encountered in trying to prove this conjecture.  One is guessing what the proper labeling of $F(\al)$ should be.  It seems that for a given $\al$ there are many labelings which work.  So narrowing the field down is where the difficulty lies.  The second problem is trying to prove that you have a CD for a certain family of fences once you have identified a suitable labeling.  Induction is the obvious proof technique, but a small change in the fence can make the lexicographic CD change in what seems to be difficult ways. 

%% Still, we are optimistic that much progress can be made on these conjectures.  We invite the reader to work on them.


%% 





%\acknowledgements{\lipsum[20]}

%% if you use biblatex then this generates the bibliography
%% if you use some other method then remove this and do it your own way
\printbibliography



\end{document}
